Author’s response to reviews

Title: Possible impact of rising sea levels on vector-borne infectious diseases

Authors:

Ranjan Ramasamy (ranjan.ramasamy@ubd.edu.bn)
Sinnathamby N Surendran (noble@jfn.ac.lk)

Version: 3 Date: 15 January 2011

Author's response to reviews:

15 January 2011

The Editor
BMC Infectious Diseases

Dear Sir/Madam,

We wish to resubmit the manuscript titled: Possible impact of rising sea levels on vector-borne infectious diseases that has been reformatted as requested as follows:

1. Authors’ contributions: We have indicated that all [both] authors read and approved the final manuscript.

2. Text: All highlighting removed

3. Title page: Affiliations and email addresses are displayed in separate sections on the title page.

4. Acknowledgements: Inserted this section directly after the authors' contributions.

5. References: All authors’ names are included in the reference list.

6. The statement of competing interests in a section titled 'Competing interests' has been made and inserted before the authors' contributions.

7. Typography: A few minor typographical changes have been made to improve presentation.

We trust that the revisions are adequate for the manuscript to be accepted and thank you for your kind consideration.

Sincerely yours,

Ranjan Ramasamy